Sheffield Hallam University

Sustainable warehouse evaluation with AHPSort traffic light visualisation and post-optimal analysis method

ISHIZAKA, Alessio, KHAN, Sharfuddin Ahmed, KUSI-SARPONG, Simonov and NAIM, Iram

Available from Sheffield Hallam University Research Archive (SHURA) at:

https://shura.shu.ac.uk/32477/

This document is the Accepted Version [AM]

Citation:

ISHIZAKA, Alessio, KHAN, Sharfuddin Ahmed, KUSI-SARPONG, Simonov and NAIM, Iram (2022). Sustainable warehouse evaluation with AHPSort traffic light visualisation and post-optimal analysis method. Journal of the Operational Research Society, 73 (3), 558-575. [Article]

Copyright and re-use policy

See http://shura.shu.ac.uk/information.html

Sustainable warehouse evaluation with AHPSort traffic light visualization and post-optimal analysis method

Alessio Ishizaka¹, Sharfuddin Ahmed Khan², Simonov Kusi-Sarpong³, Iram Naim⁴

¹ NEOMA Business School, 1 Rue du Maréchal Juin, 76130 Mont-Saint-Aignan, France, email : <u>alessio.ishizaka@neoma-bs.fr</u>

² Industrial Engineering and Engineering Management Department, University of Sharjah, Sharjah, United Arab Emirates, email: <u>skhan@sharjah.ac.ae</u>

³ Southampton Business School, University of Southampton, Highfield, Southampton SO17 1BJ, United Kingdom, email: <u>simonov2002@yahoo.com</u>

⁴ Department of Computer Science and Information Technology, Faculty of Engineering and Technology, M.J.P. Rohikhand University, Bareilly (U.P.), India, email: <u>iram.naim03cs@gmail.com</u>

Abstract - Sustainable warehousing is essential for organizations to achieve overall supply chain sustainability. Warehousing facilities have the greatest potential for reducing socio-environmental impact. Yet, both research and practice have given relatively less attention to considering all aspects of sustainability in warehouses. In order to address this gap, this study proposes combining both input from professionals and from a literature survey of triple-bottom-line theory in order to develop a sustainable warehouse criteria framework, thus contributing to sustainable organizational warehouse evaluation. The method supporting the evaluation of this framework is based on the integration of a multicriteria AHPSort-traffic light visualization technique and novel post-optimal analysis. Furthermore, the authors deployed this framework and integrated methodology in an Indian manufacturing company to evaluate and classify seven of their warehouses for decision making. The traffic light visualization technique presents and conveys the results better than numbers. Finally, the post-optimal analysis provides recommendations for efficient improvements. The findings of this study present valuable insights and guidelines for industrial managers and practitioners, especially those from the Indian manufacturing industry, for sustainable warehouse decision-making, and for improving their overall corporate sustainability performance.

Keywords: Decision analysis; Multi-criteria; Supply chain, Sustainability, AHPSort, Postoptimal analysis

1. Introduction

Sustainability, social network research and SC integration research has been evolved over the last ten years (Stolze et al. 2018; Fragoso & Figueira, 2020). Due to increasing pressure from diverse stakeholder groups, including governmental and nongovernmental agencies, customers, and employees of organizations, forcing firms to become more sustainable (Govindan et al., 2019; Sarkis, 2018, Kusi-Sarpong and Sarkis, 2019; Gunasekaran et al., 2014; Hosseini-Motlagh et al., 2019), many organizations have started to explicitly consider the triple dimension of sustainability (economic, social, and ecological) in their overall supply chain decisions (Bai, Kusi-Sarpong, Badri Ahmadi, & Sarkis, 2019; Carter & Rogers, 2008; Garetti & Taisch, 2012; P. Rao, 2002; Raut, Narkhede, Gardas, & Raut, 2017). Assessing the potential benefits of environmentally sustainable performance is challenging due to the difficulty of measuring its return on investment (García-Dastugue & Eroglu, 2019). However, it is evident from the literature that sustainable supply chain investment can increase economic performance (Kirchoff, Omar & Fugate, 2016; Hosseini-Motlagh et al., 2019). Organizations that want to achieve sustainability need to minimize tension between different conflicting factors such as short-term profitability and long-term environmental integrity (Xiao et al. 2019). Organizations are eager to fulfil customer demand in a much more effective manner. One way of achieving this goal is to have enough warehouses to hold some amount of inventory to satisfy customer demand. A very important question that confronts managers is how to identify the right warehouse operations which may represent considerable savings for the organizations, both in the medium- and the long-term (Conceição, Pedrosa, Neto, Vinagre, & Wolff, 2012). More precisely, warehouse operational decision making is one of the most important strategic decisions having a direct impact on overall

organizational supply chains (Foroozesh, Tavakkoli-Moghaddam, & Mousavi, 2018). Warehousing is one of the most critical functions in supply chains as it accounts for about 24 percent of logistical costs (European Logistics Association and Kearney, 2004).

Just as the literature on warehouse operational decisions has paid relatively less attention to environmental and social aspects in general (e.g. Xu et al., 2020; Wruck et al., 2013), organizations are also reluctant to do so in their warehousing. These organizations consider mostly economics-focused factors such as cost effectiveness and customer satisfaction as the main performance drivers (Neto, Bloemhof-Ruwaard, van Nunen, & van Heck, 2008; Linton, Klassen, & Jayaraman, 2007; Xu et al., 2020; Wruck et al., 2013). It is imperative that these organizations recognize the importance of transitioning from their current economics-based business models to sustainability-based business models (Fahimnia et al., 2015), since considering all aspects of sustainability within their warehouse operations will enhance their competitiveness (Zailani, Jeyaraman, Vengadasan, & Premkumar, 2012). Sustainable warehousing "is about integrating, balancing and managing the economic, environmental and social inputs and outputs of the warehouse operations" (Tan, Ahmed, & Sundaram, 2010). Sustainable warehouse management plays a significant part in achieving carbon efficient supply chains (Wu, Jin, Shi, & Shyu, 2017). Even though the basic function of warehouse operations is for the storage and management of goods or raw materials, yet, organizations have to comply with huge numbers of regulations and standards – ranging from the health and safety of employees, to hazardous material handling, to employees' rights – all of which have sustainability implications and consequences for the organizations (Badri Ahmadi, Kusi-Sarpong, & Rezaei, 2017). Ideally, organizations should have an effective checks and balances system in place to enable them adhere to these sustainability standards, and they should have an experienced workforce in place to implement and control such standards (Bai et al., 2019).

Several studies have investigated warehouse or facility operational decision problems considering the traditional business and economic criteria frameworks (Büyüközkan & Uztürk, 2017; Demirel, Demirel, & Kahraman, 2010; Dey, Bairagi, Sarkar, & Sanyal, 2017; Roh, Jang, & Han, 2013; Wutthisirisart, Sir, & Noble, 2015; Drezner & Drezner, 2012). A few other studies have attempted to integrate sustainability into their decisions, (see see Jha, Raut, Gardas, & Raut, 2018; He, Wang, Lin, Zhou, & Zhou, 2017; Ishtiag, Khan, & Hag, 2018; Jain & Khan, 2017; Khan, Kusi-Sarpong, Arhin, & Kusi-Sarpong, 2018; Rao, Goh, Zhao, & Zheng, 2015; Raut et al., 2017; Uysal & Tosun, 2015). Yet, none of these studies until now, have specifically considered all aspects of sustainability in their decisions and conducted investigations to aid in the classification and improvement of warehouses based on the levels of involvement in sustainability. Tools to support such decisions have also heavily relied on the analytical hierarchal process (AHP), which was developed by Saaty in 1980 (Saaty, 1980), and has since become the most widely used multi-criteria decision-making (MCDM) method. AHP takes both qualitative and quantitative inputs and breaks down complex problems in terms of different levels of hierarchy (Dweiri et al. 2016; Saaty, 1980). In the literature, several applications of AHP can be found, such as supplier selection and procurement planning (Khan, Dweiri, and Jain, 2016; Cheaitou and Khan, 2015); supply chain performance evaluation and knowledge based system development (Rehman et al. 2018; Khan, Chaabane, and Dweiri, 2019); contractor selection (Khan and Hosany, 2016); web based casting supplier assessment (Akarte et al., 2001); and forecasting method selection (Dweiri, Khan, and Jain, 2015). Similarly, it has been applied in many sustainability applications such as in hospital waste management (Ishtiaq, Khan, and Haq, 2018), supplier sustainability performance evaluation (Khan et al. 2018), reverse logistic supplier selection (Jain and Khan, 2017), and sustainable criteria selection for desalination plants (Dweiri, Khan, and Almulla, 2018). To address the current literature and practice gaps, this study proposes a comprehensive sustainable criteria framework based on a combination of literature survey and industrial

manager input from a triple-bottom-line perspective supporting sustainable warehouse classification decisions. This study further investigates this framework within manufacturing industry context in India, providing theoretical, practical, and managerial insights to support the decision-making process. The AHPSort-traffic light visualization model is utilized to support the analysis of the framework within the Indian manufacturing industry. A post-optimal analysis follows providing recommendations for improvement. Post-optimal analysis is a new technique, more advanced than traditional sensitivity analysis. Contrary to sensitivity analysis, it does not only vary the scores of alternatives to assess the robustness of the classification, but it also searches for the cost-efficient variation of the scores to upgrade by one level.

This study targets India and its manufacturing industry for some key reasons, some of which are discussed as follows. The country is part of BRICS nations (part of the major emerging economy nations), which are known to have significant influence on regional affairs and are part of the G20 (BBVA Annual Report, 2012). The manufacturing industry happens to be one of the fastest growing industries in India with potential growth/revenue hitting some US\$1 trillion in 2025 (Mehta & Rajan, 2017). Unfortunately, this industry's increasing growth does not match that of technological improvement and organizational practice, hence investment has been limited. There is also a need to improve sustainable performance in the Indian manufacturing industry, as growth will be hampered unless considering all aspects of sustainability. An important starting point to achieve this goal is to ensure that warehouse operations are sustainable by introducing sustainable criteria and practices into warehouse decision making. This study aims to help such initiatives.

The objectives of this paper are to:

a) Propose a sustainable criteria framework based on a combination of literature survey and industrial manager inputs and from a triple-bottom-line perspective.

- b) Introduce the AHPSort method coupled with a traffic light visualization technique and post-optimal analysis to aid in the evaluation and classification of sustainable warehouses and propose improvements.
- c) Evaluate and classify, from an Indian manufacturing company context, levels of sustainability involvement in selected warehouses using the sustainability criteria framework and the integrated method.
- d) Provide recommendations on how to improve sustainability from a cost-efficient approach.

The paper makes the following four contributions:

(1) It identifies and develops a multi-level sustainable criteria framework as a theoretical basis and decision-making reference for manufacturing sector sustainable warehouse classification decisions;

(2) It proposes AHPSort method coupled with a traffic light visualization technique as a decision support tool for aiding the evaluation and classification of sustainable warehouses and other facilities;

(3) It develops a novel post-optimal analysis method for recommending costefficient improvements.

(4) It applies this methodology using empirical data in the context of the Indian manufacturing sector.

The remainder of the paper is organized as follows: Section 2 provides a brief literature survey of sustainable organizations, warehousing, warehousing sustainability criteria, and MCDM tools for supporting sustainable warehousing decisions. Section 3 describes the methodology composed of the AHPSort-traffic light visualization method and post-optimal analysis. Section 4 presents the case study. Section 5 discusses the results, and finally Section 6 presents the conclusion and some future directions.

2. Literature Background

2.1. Sustainable organization

Over the last decade, the concept of sustainability has gained much attention from both supply chain researchers and practitioners (Badri Ahmadi et al., 2017; Bai, Kusi-Sarpong, & Sarkis, 2017; S Kusi-Sarpong, Varela, Putnik, Ávila, & Agyemang, 2018; Song & Li, 2019). This wave of interest in sustainability by practitioners is a result of the growing awareness of businesses and communities at large of the need to transition their business operations to sustainable business operations (Perrott, 2014; Xing & Dangerfield, 2011). This initiative, however, can be advance through the integration of sustainability concerns into organizational and business operations (Mežinska et al., 2015; Sarkis, 2018). Sustainability has been defined in terms of its economic, social, and environmental dimensions (Carter and Rogers 2008). Therefore, organizations must balance economic activities with social and environmental concerns to achieve sustainability (Badri Ahmadi et al., 2017; Bilbao-Terol et al., 2018). In this way, organizations must manage their operational activities to maximize *economic gains*, improve *stakeholders' wellbeing* and reduce *negative environmental impact* in order to achieve a truly sustainable organization (Gupta et al., 2020).

Yet, organizations will need to demonstrate innovativeness (Orji et al., 2019; Gupta et al., 2020; Ji, & Gunasekaran, 2014) in order to integrate sustainability in their business operations. The organizations pursuing innovation initiatives to enhance sustainability need to introduce new processes, or modify organizational practices, systems, and techniques, to minimize environmental and social impact, and thereby increase competitive advantage (Beise & Rennings, 2005). For example, organizations can develop innovative strategies such as using renewable energy sources to power their warehouses to deal with and enhance sustainability performance in their warehouses (Orji et al., 2019; Cai and Zhou, 2014). Sustainable innovations thus can benefit organizations in several ways, including by reducing costs, and improving the image and profit of the

organizations (Aguado et al., 2013). The importance of sustainable innovation in moving to organizational sustainability cannot be underestimated (Kusi-Sarpong at al., 2019). By extension, organizations need to innovate for sustainability in order to enhance their operations performance (Varadarajan, 2017).

2.2. Sustainable Warehousing

Warehousing includes storage, retrieval operations, organizing, material handling equipment, media for the storage of materials, and the necessary building facilities to protect goods (Gunasekaran et al., 1999). A warehouse is therefore an important part of organizational operations (da Silva et al., 2015). This is because it connects the production lines to the customers and plays an important role in delivery performance (Gunasekaran et al., 2001). Therefore, improving the effectiveness of warehouse operations can lead to increased overall organizational performance (Gunasekaran et al., 1999). However, warehouse operations are confronted with numerous sustainability issues and organizations have to make certain critical decisions to address these concerns. Unfortunately, most warehouse decisions have predominately been focused on economic sustainability, such as minimizing costs or maximizing customer service level (Rao et al., 2015). Thus, limited studies have attempted to investigate the sustainability of warehouse operations. With the emphasis on corporate social responsibility and environmental awareness, this goal has grown to include the consideration of other criteria to help address the challenge of current sustainability requirements of warehousing decisions. A wrong sustainable warehouse decision may result in additional costs for the organization, such as fines, while making no impact on the level of customer service (Conceição et al., 2012). Therefore, organizations must not only consider the economic factors, but also social and environmental factors, when ensuring that their warehouse operations and operational environments are not negatively affecting their sustainable supply chain performance (Tan et al., 2010).

While considering sustainability issues in supply chains, previous studies have paid more attention to the investigation of issues such as procurement (Walker and Brammer, 2012), transportation (Schneider, 2013), and manufacturing (Garetti & Taisch, 2012) . Studies that focus explicitly on sustainable warehousing are limited. Supply chain activities in warehouse operations may have severe environmental and social problems with serious economic consequences (Kusi-Sarpong, Sarkis, & Wang, 2016). The negative impact of warehouse facilities and activities has grown exponentially, threatening and increasing the burden on local communities (Yuan, 2018). Organizations must therefore lead in this transition towards sustainable technologies and practices by implementing environmentally sound and socially responsible warehousing operational practices (Abdallah, Diabat, & Simchi-Levi, 2012). Given the importance of addressing a host of environmental and social problems inherent in warehousing operations (Pagell & Shevchenko, 2014), sustainable warehousing is key solution for alleviating the negative sustainability impact and creating beneficial consequences (Zailani et al., 2012).

In addition to traditional economic and business issues, sustainable warehousing decisions consider environmental and social issues when aiming to achieve higher sustainable performance. Indeed, many organizations have focused on the use of energy-efficient systems in their warehouse operations (Meneghetti & Monti, 2013). Marchant, (2010) proposed a three-stage sustainable warehouse model that deals with the business, economic, environmental, and social aspects of warehousing, and attempted to capture a wider range of criteria to aid organizations in achieving minimum negative sustainability impact. In another study, Tan et al., (2010) presented a discussion on the concept of sustainable enterprise simulation models from the perspective of a warehousing and distribution company, explaining the interconnectivity among the various sustainability dimensions in practice. Similarly, Żuchowski, (2015) argued that strategic implementation of sustainable solutions for warehouse management minimizes greenhouse gas emissions and resource consumption and, in the long term, leads to sustainable warehousing.

Overall, most companies' sustainable warehousing decisions contributing to such a transition are principally driven by compliance with regulation. Yet, the lack of understanding of sustainable regulations and its implications on the operations of warehousing could be challenging to achieve the sustainability goal within the warehouse sector.

2.3. Warehousing sustainability criteria

In order to evaluate warehousing operations, many criteria must be taken into consideration. A number of studies have considered different qualitative and quantitative criteria, such as governmental policies and regulations (Dashore & Sohani, 2013; Demirel et al., 2010; Ketikidis, Hayes, Lazuras, Gunasekaran, & Koh, 2012; Luthra, Kumar, Kumar, & Haleem, 2011); climatic conditions (Ashrafzadeh, Rafiei, Isfahani, & Zare, 2012; Dey et al., 2013); technology availability (Amjed & Harrison, 2013; Ashrafzadeh et al., 2012; Karmaker & Saha, 2015); availability of a sustainable skilled workforce (Amjed & Harrison, 2013; Karmaker & Saha, 2015; Luthra et al., 2011; Luthra, Qadri, Garg, & Haleem, 2014); provisions for energy saving (Amjed & Harrison, 2013; Diabat & Govindan, 2011; Dubey, Bag, Ali, & Venkatesh, 2013; González, Sarkis, & Adenso-Díaz, 2008). From the above literature, it is clear that many frameworks for warehouse evaluation have been proposed, but are mostly limited to economic and business criteria, with a few focusing on environmental sustainability. While broader sustainability consideration has begun to emerge (see e.g. Uysal & Tosun, 2015), these frameworks do not examine sustainability concerns. Taking the triple-bottom-line theoretical view, this study introduces a unified and more comprehensive sustainable criteria framework for helping organizations evaluate sustainable warehousing operations.

In this study, through an initial literature survey, we collected and tabulated various criteria pertaining to sustainable warehousing decisions. We then sent this to seven experts (listed in Table 2, in Section 4) for review, validation, and categorization, as either social, economic, or environmental factors, and integration into a single and unified

framework to aid the evaluation and classification of sustainable warehousing. This process was reviewed and completed in a group and facilitated by one of the authors via videoconferencing. Table 1 depicts the final list of criteria and their triple-bottom-line categorization as being either social, environmental, or economic, along with a brief description of each criterion according to the expert review group.

No.	Dimension	Criteria	Brief Description	References
		Inventory Costs (IC)	Includes the cost required to hold and maintain the inventory in physical terms, the ordering cost needed to prepare and process the order, and associated labour cost for arranging inspection of the goods.	(Cagliano, Demarco, Rafele, & Volpe, 2011; Gallmann & Belvedere, 2011; Rimiene, 2008)
		Labor Costs (LC)	Cost incurred towards salary or wages of the manpower mobilized for the related work. May be on a daily, weekly, monthly basis, or on the basis of the number of jobs performed. Also includes other benefits and taxes as applicable.	(Cagliano et al., 2011)
1	Economic	Costs as a % of Sales (CAS)	Direct expense cost involved to generate finished goods. Includes the cost of raw materials, processing, and manpower costs, etc. which are required for the production of goods.	(A, N. Subramanya, & M. Rangaswamy, 2012)
		Maintenance Costs (MC)	Cost required to keep the infrastructure of a factory operational.	(de Marco & Mangano, 2011; Andrew Johnson & McGinnis, 2011; Andy Johnson, Chen, & McGinnis, 2010; E. Johnson, 2008)
		Cost of Land (CoL)	Cost incurred by any company in acquiring the land on which its entire factory is to be established, as well as cost required to ready it for use.	(Ashrafzadeh et al., 2012; Demirel et al., 2010; Glasmeier, 1996; Sivitanidou, 1996;

Table 1: Sustainability Criteria for Warehouse Evaluation and ClassificationDecisions

				Turğut, Taş, Herekoğlu, Tozan, & Vayvay, 2011)
		Overhead Costs (OHC)	Indirect and fixed costs not related to direct manufacturing costs. Includes expenditures incurred on travel, insurance, legal fees, rents, utilities, etc.	(Lambert, Stock, & Ellram, 1998; Tompkins & Smith, 1998)
		Construction Costs (CC)	Cost to company for an entire project involved in construction work for capital improvement.	(Lambert et al., 1998; Tompkins & Smith, 1998)
		Renewable Energy Sources (RES)	Natural sources of energy, such as solar energy. It signifies the quantum of natural sources of energy used by the company	(Marchant, 2010)
		Daylight Usage (DU)	Suggests the utilization of daylight freely available in daytime.	CarbonTrust. (2007)
		Noise pollution (NP)	Noise generated by running of machinery in plants and propagated in the environment. A high level of noise is harmful and treated as noise pollution.	(Gazeley, 2004)
2	Environmental	Recycling Facility (RF)	It refers to the company's available facilities for reusing abandoned material or generated waste.	(NZBCSD, 2003)
		Material Handling Equipment Power Sources (MHEPS)	Defines the company's available equipment or facilities for handling of material and power sources, such as a tipper, forklift, etc.	(Buckley, H. 2006; Department for Transport (DfT), 2006)
		Artificial Lighting Scheme (ALS)	The arrangement for illumination which consumes electricity.	(Carbon Trust, 2007; Marchant, 2010)
		Temperature Control (TC)	The arrangement provided to maintain and control the temperature of concerned space.	(Marchant, 2010)
		Aisle Design (AD)	Indicates the design and width of walkways in production blocks and other storage areas of the factory.	(Hassan, 2002)
		Work-Life Balance (WLB)	The equilibrium required to balance professional and personal lives.	(Hudson, 2005)
3	Social	Shift Roster (SR)	A written message which shows the timing of daily work for a particular working group.	(New Zealand Department of Labor (NZDoL), 2007)
		Driver or Operator Training (DT)	A focused and dedicated training program for operators, drivers, or any other worker for their specific	(Ward <i>et al.</i> 2004)

		job, to enhance their knowledge and	
		enable them to perform their work	
		effectively.	
		The system by which the company	
		monitors employee health and	(Kuorinka, Lortie,
		ensures their safety in the workplace.	& Gautreau, 1994;
	fearth and	Generally, a dedicated company	Larsson &
30	alety (OHS)	group or department carries out all	Rechnitzer, 1994)
		the activities related to OHS.	
Picki	ing Processes	The company's system for finding or	(Tomking 1006)
	(PP)	extracting products.	(TOTIKITIS 1990)
		The basic facilities offered to	
Wel	fare Facilities	employees, like clean drinking water,	(Australia, S.,
	(WF)	meals, toilets, protective equipment,	2011)
		etc., essential in the workplace.	
Emo	raanay Poom	A dedicated space in a particular	
Ellie		building or department for	(Victoria, W, 2008)
		employees' case of any emergency.	

This framework was then used in a case study setting to evaluate and classify sustainable warehousing in section 4.

2.4. MCDM methods for supporting Warehouse Decisions

Warehouse sustainability evaluation is a multi-criteria decision-making (MCDM) involving several conflicting criteria and choices requiring adequate methods (Kusi-Sarpong, Sarkis, & Wang, 2016; Aguezzoul, 2014; Orji et al., 2020). Such criteria may be both qualitative and quantitative (Khumawala & Clay Whybark, 2008; Magee et al., 1985). In the literature, several studies have used different MCDM methods for warehouse operations evaluation, such as integrated fuzzy and TOPSIS-SAW-MOORA (Dey, Bairagi, Sarkar, & Sanyal, 2016) AHP –TOPSIS (Özcan, Elebi, & Esnaf, 2011); FAHP (Khan, Dweiri, & Chaabane, 2016); hybrid fuzzy MCDM approach (Dey, Bairagi, Sarkar, & Sanyal, 2013); AHP with hybrid fuzzy and TOPSIS (Karmaker & Saha, 2015); AHP (García et al., 2014; Korpela & Tuominen, 1996). The MCDM methods employed in those studies aid in the evaluation and selection of sustainable warehouses but are unable to assist in classifying sustainable warehouses. For the first time, therefore, this study introduces AHPSort-

Visualization and post-optimal analysis method – an MCDM model for sustainable warehouse evaluation enabling evaluation, classification, and recommendations for improvement.

3. Methodology

3.1. AHPSort-based Traffic Light Visualization method

AHPSort is an extension of AHP (Saaty, 1977) for sorting problems. It was invented in 2012 (Ishizaka, Pearman, & Nemery, 2012), and has since been applied to several case studies (Krejčí & Ishizaka, 2018; Lolli, Ishizaka, & Gamberini, 2014; López & Ishizaka, 2017; Miccoli & Ishizaka, 2017). This section describes the main steps of AHPSort.

First, the problem needs to be structured. In the previous papers, the problem was defined with the goal of the problem, the criteria c_j , j = 1,..., m, and the alternatives a_k k = 1,..., l. In this paper, we structure the problem as a hierarchy, where sub-criteria are added. This allows to be more precise and to identify better the fine performance of the alternatives. As it is a sorting problem, we need also to define the classes C_i , i=1,...,n, where n is the number of classes. The classes are ordered and have a label (e.g., excellent, good, medium, bad). These classes need to be defined. This is done generally with local limiting profiles lp_{ij} , which indicate the minimum performance needed for each criterion j to belong to a class C_i . The classification is then performed as described in (Ishizaka, Pearman, & Nemery, 2012).

AHPSort has many advantages, especially from the practical point of view. The decision-maker is asked to conduct pairwise comparisons. This means that the decision-maker focuses on only two elements at the time, which is more precise and less confusing than directly evaluating elements (Millet, 1997; Por and Budescu, 2017). It does not require difficult inputs to elicit on like preference and indifference threshold or technical parameters as in Electre-Tri and ELECTRE-SORT (Almeida-Dias et al., 2010; Ishizaka and Nemery, 2014) and FlowSort (Nemery and Lamboray, 2008).

Moreover, AHPSort requests a redundancy of information which allows to check the consistency of the responses. If the decision-maker is not consistent with the responses to pairwise comparisons, the decision-maker is asked to revise those inputs/responses until consistency is achieved.

To better illustrate the results, this study introduces and integrates the traffic light visualization approach with the AHPSort method. The traffic light method is a visual method to display the classification of the alternatives in relation to the goal, criteria and sub-criteria. It is an effective way to communicate performance information. The indication of performances is done using the three colours of the real traffic lights (red, orange and green). The number of variation of colours can be extended with the number of classes. Good and poor performances can easily be identified, and appropriate action can be taken to improve performances. The traffic light visualization approach is an extension of the AHPSort method and offers a finer understanding of the classification of the alternatives, in this case, the warehouses.

3.2. Post-optimal analysis

Researchers have long recommended using sensitivity analysis after applying MCDM method in order to integrate uncertainties (Saaty, 1980). The aim of the exercise is to determine the stability of the ranking. Barron and Schmidt (1988) proposed a new type of sensitivity analysis, where they determined the minimum weight modification in order that an alternative become equal to or be exceeded by Δ for a specific alternative. They applied it to the additive multi-attribute value method. Wolters and Mareschal (1995) proposed the same analysis for PROMETHEE. Triantaphyllou and Sánchez (1997) searched for the most critical criteria and alternative performances where the smallest change altered the existing ranking for the weighted sum model, the weighted product model, and the analytic hierarchy process. Later, the minimum performance modification for improving the score of an alternative, in order that it be weakly preferred over another,

was applied to PROMETHEE and the weighted sum method (Hyde and Maier, 2006; Hyde et al., 2005). Recently, Ciomek et al. (2018) and Kadziński et al. (2016) have proposed a post-factum analysis. It assesses the consequences of different variations for the additive value model. In particular, it quantifies the possible and necessary improvements that need to be made to reach a certain target. Both studies also estimated the potential deterioration that could be afforded, whilst keeping the same target. The same analysis has also been proposed for TOPSIS (Dutta et al., 2019). However, it should be noted that the cost of improving one alternative is not necessarily the same as for another alternative. In this paper, we propose a post-optimal analysis, where the improvement costs of each alternative are taken into account. Our approach is applied on AHPSort, but could also use other sorting or ranking techniques.

The aim of the post-optimal analysis is to identify the minimal improvements on the performances that could upgrade the performance of action a_i to a higher level. This recommendation should be formulated in order to minimize costs.

Formally, suppose an action a_i is assigned to a class C_{j} , and action a_i is denoted a_i^+ after the improvement of its performances have been operated, such that:

$$f_j(a_i^+) = f_j(a_i) + \Delta_j^i \quad \forall j$$
(5)

where $\Delta_j^i \ge 0$ represents the increase in performance of action a_i on criterion f_j .

In order to be assigned to a better category C_{j-1} , we need to achieve the following condition (2):

$$lp_{j-1} > f(a_i^+) \ge lp_{j-2}$$
 (6)

In addition, the improvements on the performances of action a_i must generally meet the following two constraints of (3) and (4).

Performances cannot be decreased:

$$f_j(a_i^+) \ge f_j(a_i) \text{ or } \Delta_j^i \ge 0 \quad \forall i, j$$

$$\tag{7}$$

• Improvements on performances have an upper limit:

$$U_{j} \geq f_{j}(a_{i}^{+}) \text{ or } U_{j} - f_{j}(a_{i}) \geq \Delta_{j}^{i} \geq 0 \quad \forall i, j$$
where, U_{j} is the upper limit on criteria j
$$(8)$$

Additional feasibility constraints may be added to the model: for example, that some improvements on a precise criterion are not possible.

We are thus searching for the values Δ_j^i such that the generated costs $C(\Delta_j^i)$ are minimum, whilst satisfying the constraints (6), (7), and (8). This problem can be solved with the following mathematical program:

$$\begin{array}{l} \operatorname{Min} \sum_{j=1}^{n} \mathbf{C}(\Delta_{j}^{i}) \\ \text{s.t.} \\ f(a_{i}^{+}) \geq l p_{j-2} \\ \Delta_{j}^{i} \geq 0 \\ U_{j} \geq f_{j}(a_{i}^{+}) \end{array}$$

$$\begin{array}{l} (9) \\ \end{array}$$

4. Case study

A case study is used to aid in the investigation and classification of the sustainable warehouse. The study uses industrial managers from an Indian manufacturing company. These managers were involved in the evaluation and classification processes of the warehouses based on their involvement levels in all aspects of sustainability. We followed a combination of purposeful and self-selection sampling approaches to choose these companies' respondent managers. We present the details of the Case Study Company and warehouses in sub-section 4.1, and the respondent managers involved in the decision in Table 2.

Expert #	Position	Working Experience (in years)
1	Assistant General Manager	33
2	Assistant Engineer	13
3	Senior Engineer	7
4	Assistant Engineer	9
5	Assistant Engineer	3
6	Engineer	6
7	Assistant Engineer	15

Table 2: Case Company's Respondent Managers involved in the decision

4.1 Case problem description – XYZ Company Ltd.

India, the case country for this study, is one of the BRICS nations and is at a relatively immature stage in terms of sustainable development implementation (Mani et al., 2016). The manufacturing sector is especially young with respect to sustainable development (Kusi-Sarpong, Gupta, & Sarkis, 2019; Kusi-Sarpong et. al, 2019).

We now provide an overview of the case company under consideration. The case company, henceforth referred to as XYZ Company Ltd., is an industrial organization dealing in power plant components. XYZ Company Ltd. is a manufacturer of large-size casting and forging components of various types of steel, such as alloy, creep-resistant, and supercritical grade steels. XYZ Company Ltd. is a pioneer company forging ahead to make the heaviest castings and forgings in India. XYZ Company Ltd. produces steel products in the order of 10,000 metric tons (MT) per year. The manpower of XYZ Company Ltd. is around 1500, including top management, working executives and other elements of the workforce. XYZ Company Ltd.'s operational processes starts from engineering to design, manufacture and supply through its labor and well versed with all engineering equipment. It also has strong vendor base throughout the country and ancillary units nearby to its location.

XYZ Company Ltd. has planned to improve its sustainability performance at their warehouses due to series of strike actions both by its employees, because of poor work

conditions, and local activists in respond to the negative health and environmental impacts. Since warehouses are mostly located away from the manufacturing points and are closer to customers (Schmitt et al., 2015), XYZ Company Ltd. is keen to identify warehouses with higher performance in all aspects of sustainability to enable them improve their lower-performing warehouses. In doing so, XYZ Company Ltd. has initiated the decision-making process by selecting those warehouses that have received huge complaints from both employees and customers or local activists. XYZ Company Ltd. wants to categorize its existing warehouses in terms of levels of involvement in sustainability. In order to do this, they established four categories, including excellent, good, medium, and bad, to help them identify the levels of involvement and how much effort would be required to improve upon low-performing warehouses.

4.2 Problem structuring

The list of collected criteria from the literature (Table 1) was presented to the company managers (Table 2). They approved all criteria, apart from "Costs as a % of sales" and "Temperature control". They deleted these two criteria because they were not related to them in exact terms: they did not measure costs in terms of sales percentages, and they did not have any mechanism for measuring temperature control. In order to measure cost, they proposed that "cost preorder" be added to the list, as this is their main parameter for measuring the economic viability of their warehouses. Similarly, they also added transportation costs for the same reason. We then asked them to define the limiting profile of the classes. After some discussion between the managers, they agreed on the values shown in Table 3.

Sustainability Dimension	Criteria	Limiting Profile Excellent	Limiting Profile good	Limiting Profile Medium
Economical	Inventory Costs (IC)	10% of total costs	15% of total costs	20% of total costs

Table 3: Limiting profiling

	Labor Costs	5% of gross	8% of gross sales	11% of gross sales
	(LC)	sales		
	Maintenance Costs	8% of	12% of	16% of
	(MC)	manufacturing	manufacturing	manufacturing
	(((())))	costs	costs	costs
	Cost of Land (CoL)	\$10/sq ft	\$13/sq ft	\$16/sq ft
	Overhead Cost (OHC)	20% of COGS	25% of COGS	30% of COGS
	Construction Cost (CC) of		\$17/sq ft	\$19/sq ft
	reinforced cement concrete (RCC)	\$15/sq ft		
	Cost per order (C/O)	\$12/pallet	\$15/pallet	\$18/pallet
	Transportation cost (TC)	7% of total	9% of total	11% of total
	Transportation cost (TC)	revenue	revenue	revenue
	Renewable Energy Sources	50% of total	40% of total	30% of total
	(RES)	energy usage	energy usage	energy usage
	Daylight Usage	90% of WH	85% of WH	80% of WH
	(DU)	activities	activities	activities
	Noise pollution (NP)	55 dB (A)	65 dB (A)	80 dB (A)
Environmental	Recycling Facility (RF)	5 facility	4 facility	3 facility
	Material Handling	EOO/ MUEDS on	35% MHEPS on	25% MHEPS on
	Equipment Power Sources (MHEPS)	solar	solar	solar
	Artificial Lighting Scheme	12 footcandles	18 footcandles	24 footcandles
	(ALS)	(fc)	(fc)	(fc)
		Pallet per Sq	Pallet per Sq	Pallet per Sq
	Aisle Design (AD)	metre ratio	metre ratio	metre ratio
		1	1.1	1.2
	Work-Life Balance (WLB)	0 late sittings	1 late sitting	2 late sittings
	Shift Bostor (SD)	1 night	2 night	3 night
		shift/month	shifts/month	shifts/month
Cosial	Driver or Operator Training (DT)	4	3	2
SOCIAI	Occupational Health	Asceroditation	3 accreditation	2 accreditation
	and Safety	4 accreditation	certificates	certificates
	(OHS)	Certificates		
	Dicking Processos (PD)	55% automated	45% automated	35% automated
		processes	processes	processes
	Welfare Facilities (WF)	7	6	5
	Emorgonay Room (EP)	4 fully equipped	3 fully equipped	2 fully equipped
		rooms	rooms	rooms

4.3 Criteria weighting.

The mangers were asked to evaluate the criteria based on pairwise comparisons. As some matrices were quite large – as in 8x8, inconsistencies were inevitable. To decrease them,

we used the Saaty's (2003) interactive method. The least consistent entry is highlighted and the decision-maker is invited to reassess it. This process is repeated with the next inconsistent entry, and so on, until the matrix consistency is below the acceptable consistency threshold of 0.1. The weightings resulting from these comparisons are given in Table 4. The social and environmental dimensions of sustainability seem the most important, with weights measuring more than 80% of total weight. However, the economic dimension contributes to only 16.8% of the overall weight of importance. This shows that the company's experts strongly believe that in order to achieve overall sustainability in their supply chain function, they have to pay more attention to the environmental and social aspects of their warehouse operational decisions.

Sustainability Dimensions	Criteria	Weights
	Inventory costs (IC)	0.080
	Labor costs (LC)	0.166
	Maintenance costs (MC)	0.150
Economic	Cost of Land (CoL)	0.090
(0.168)	Overhead Costs (OHC)	0.123
	Construction Costs (CC)	0.082
	Cost per order (C/O)	0.083
	Transportation costs (TC)	0.225
	Renewable Energy Sources (RES)	0.113
	Daylight Usage (DU)	0.079
Environmental	Noise pollution (NP)	0.321
(0.349)	Recycling Facility (RF)	0.236
	Material Handling Equipment Power Sources (MHEPS)	0.150
	Artificial Lighting Scheme (ALS)	0.101
	Aisle Design (AD)	0.035
	Work-Life Balance (WLB)	0.091
	Shift Roster (SR)	0.054
Social	Driver/Operator Training (DT)	0.340
(0.484)	Occupational Health and Safety (OHS)	0.171
	Picking Processes (PP)	0.147
	Welfare Facilities (WF)	0.129
	Emergency Room (ER)	0.033

Table 4: Weightings of the criteria

4.4 Classification

Each manager was asked to evaluate the warehouses against the limiting profiles (see Figure 1).

Classification of Alternatives																		
Circle one (1 = Equal 2, 4, 6, 8 are	num 3 = 1 e inte	ber (Mode	per i erate diate	row 5 valu	belo = St ies	ow u trong	sing 7	the = Ve	ery s	le: tron	g 9) = E	xtre	ne				
		the	malat	ius e	orfor			f the	Com	tral			inct	the I				
Com Inventory Co	pare	the 8%	reiau	ive p	eno	Indi	ice o	rune	Cen	u di s	SLOFE	: aga	Inst	ine i	nvei	ntor	y co	st criterion
Inventory Co Civil Store	pare st = 9	8%	7	6	5	4	3	2	1	2	3	4	5	6	7	8	9 CO	Excellent ip
Inventory Co Civil Store Civil Store	pare st = 9 9	8% 8	7	6 6	5	4	3	2 2	1	2	3	4 4	5	6	7 7	8	9 9	Excellent Ip Good Ip
Inventory Co Civil Store Civil Store Civil Store	pare ost = 9 9 9	8% 8 8 8	7 7 7	6 6	5 5 5	4 4 4	3 3	2 2 2	1 1	2 2 2	3 3	4 4 4	5 5 5	6 6 6	7 7 7	8 8 8	9 9 9	Excellent lp Good lp Medium pl
Inventory Co Civil Store Civil Store Civil Store Excellent Ip	pare 9 9 9	8% 8 8 8 8	7 7 7 7 7	6 6 6	5 5 5 5	4 4 4 4	3 3 3 3	2 2 2 2	1 1 1	2 2 2 2	3 3 3	4 4 4 4	5 5 5 5	6 6 6	7 7 7 7 7	8 8 8 8	9 9 9 9	Excellent ip Good ip Medium pl Good ip
Inventory Co Civil Store Civil Store Civil Store Excellent lp Excellent lp	pare st = 9 9 9 9	8% 8 8 8 8 8 8 8	7 7 7 7 7 7	6 6 6 6	5 5 5 5 5	4 4 4 4 4	3 3 3 3 3	2 2 2 2 2	1 1 1 1	2 2 2 2 2 2	3 3 3 3	4 4 4 4	5 5 5 5 5	6 6 6 6	7 7 7 7 7 7	8 8 8 8 8	9 9 9 9	Excellent Ip Good Ip Medium pl Good Ip Medium Ip

Figure 1: Extract from the questionnaire

After calculating the priorities with Equation (1) and aggregating them with Equations (2) and (3), we found that most warehouses are classified as "good" with only two as "medium" and these can be found in Table 5.

Warehouse	Alternative	Excellent lp	Good Ip	Medium lp	class
1	0.289	0.485	0.169	0.057	Good
2	0.216	0.470	0.212	0.101	Good
3	0.281	0.482	0.176	0.060	Good
4	0.284	0.452	0.192	0.071	Good
5	0.188	0.524	0.214	0.074	Medium
6	0.170	0.544	0.216	0.070	Medium
7	0.215	0.518	0.196	0.071	Good

Table 5: Classification of warehouses

To have a finer understanding of the classification of the warehouses, we used a traffic light approach (Figure 2). This visualization approach provides better feedback. For example, Warehouse 1's excellent economic sub-criteria qualifies as "maintenance cost"

because it is well within the defined excellent limiting profile. Similarly, "aisle design" and "picking process" in the warehouse are the excellent social sub-criteria of Warehouse 1, and "artificial lightening system" is the excellent environmental sub-criteria of Warehouse 1. The bad sub-criteria of Warehouse 1 includes economic "cost per order", social "occupational welfare and health and safety", and environmental "renewable energy sources" and "recycling facilities". The same analysis can be done for the other warehouses. This traffic light approach helps managers and decision-makers specifically evaluate the sustainability of each warehouse and put together the necessary improvement strategies and plans.

Warehouse 1 (civil store)

Economic											Soc	cial					En	viror	nmer	ntal	
IC LC MC CL OC CC CO TC								AD	WLB	SR	DT	OH	PP	WF	ER	RE	DU	NP	RF	MH	AL

Warehouse 2 (Instrument store)

Economic											Soc	ial					En	viror	nmer	ital	
IC LC MC CL OC CC CO TC								AD	WLB	SR	DT	OH	PP	WF	ER	RE	DU	NP	RF	MH	AL

Warehouse 3 (Central store)

Economic											Soc	cial					En	viror	nmer	ntal	
IC LC MC CL OC CC CO TC								AD	WLB	SR	DT	ОН	PP	WF	ER	RE	DU	NP	RF	MH	AL

Warehouse 4 (Electrical maintenance store)

Economic							Social							Environmental							
IC	LC	MC	CL	OC	CC	CO	ТС	AD	WLB	SR	DT	OH	PP	WF	ER	RE	DU	NP	RF	MH	AL

Warehouse 5 (Production 1)

Economic						Social						Environmental									
IC	LC	MC	CL	OC	CC	CO	TC	AD	WLB	SR	DT	OH	PP	WF	ER	RE	DU	NP	RF	MH	AL

Warehouse 6 (Production 2)

Economic							Social						Environmental								
IC	LC	MC	CL	OC	CC	CO	TC	AD	WLB	SR	DT	OH	PP	WF	ER	RE	DU	NP	RF	MH	AL

Warehouse 7 (Mechanical maintenance)

Economic							Social							Environmental					
IC	LC	MC	CL	00	CC	CO	тс	AD	WLB	OH	PP	WF	ER	RE	DU	NP	RF	MH	AL
Excellent				Good			Medium				Bad								

Figure 2: Traffic light classification by criteria of the warehouses

4.5 Post-optimal Analysis

We ran a post-analysis for all warehouses. The goal was to upgrade to the next level. The first step was to ask the general manager of each warehouse (W1-W7) to provide us with the cost of improving the score of each criterion for his warehouse by 1%. As the general manager may not have been aware of all costs, he also met with the finance manager to give a best estimate. As various warehouses store different types of inventory, costs vary accordingly. While the number of employees working in these warehouses may be the same, their cost is different. For example, in Warehouse 1 (civil) and Warehouse 2 (instrumentation), there are significant differences in the cost of handling the material through human intervention, and thus manpower cost is different for these warehouses. This is because the civil store consists of large and heavy items compared to the instrumentation store which, mainly stores small but delicate equipment. Some warehouses, like those of electrical maintenance and production, function around the clock, whereas others, like central and civil warehouses, are open only during the day. Similarly, for different warehouses, their maintenance cost compared to other economic, environmental, and social factors vary according to the nature and use of each warehouse. Therefore, the cost of improving the warehouse also varies according to the nature, size, and functioning of its stock. Table 6 presents an estimate of such costs. It may be noted that four criteria cannot be improved in the current situation.

No	Criteria			Cost o	f improve	ment		
		W1	W2	W3	W4	W5	W6	W7
1	Inventory Cost	\$180	\$50	\$450	\$120	\$250	\$325	\$180
2	Labour Cost	\$2	\$1	\$5	\$1	\$3	\$4	\$2
3	Maintenance Cost	\$24	\$12	\$32	\$30	\$50	\$55	\$28
4	Cost of Land		Ν	lo cost as	it will be	same land		
5	Overhead Cost	\$32	\$12	\$14	\$20	\$40	\$46	\$20
6	Construction Cost	\$12	\$11	\$14	\$14	\$18	\$21	\$10
7	Cost per Order	\$8	\$6	\$15	\$10	\$17	\$18	\$11
8	Transportation Cost	\$4	\$2	\$4	\$2	\$6	\$7	\$4
9	Renewable Energy Sources	\$220	\$60	\$650	\$400	\$500	\$502	\$280
10	Daylight Usage	\$15	\$23	\$45	\$25	\$35	\$36	\$25
11	Noise Pollution	\$1,50	\$1	\$7	\$2	\$3,50	\$3,55	\$2
12	Recycling Facility	\$135	\$80	\$450	\$130	\$300	\$305	\$200
13	Material Handling Equipment							
	Power Sources	\$65	\$30	\$150	\$70	\$100	\$99	\$45
14	Artificial Lighting Scheme	\$2	\$2	\$25	\$2	\$5	\$6	\$4
15	Aisles Design		Not	able to c	hange the	aisle desi	gn	
16	Work-Life Balance			Not a	ble to imp	orove		
17	Shift Roster	\$240	\$230	\$220	\$200	\$450	\$445	\$200
18	Driver/Operator Training	\$2	\$2	\$7	\$4	\$9	\$8	\$5
19	Occupational Health and							
	Safety			Not a	ble to imp	orove		
20	Picking Processes	\$2	\$2	\$6	\$3	\$4	\$5	\$2
21	Welfare Facilities	\$45	\$70	\$46	\$55	\$90	\$93	\$52
22	Emergency Room	\$12	\$15	\$30	\$20	\$25	\$22	\$13

Table 6: Cost of improving 1% of sustainable performance

The post-analysis improvement is calculated with the mathematical program of Equation (9). Where it is not possible to improve criteria, we use very large costs.

Table gives the optimal improvements. It may be noted that some criteria do not contribute to optimal improvement of the warehouse: renewable energy sources, recycling facility, material equipment handling, power sources, and the shift roster. This is not surprising, as they are the most expensive criteria to improve (Table 6). The highest improvement effort lies in the costs of labour, noise pollution, and picking processes, which are cheap to improve (Table 6).

No.	Criteria	W1	W2	W3	W4	W5	W6	W7
1	Inventory Costs	0 %	0 %	0 %	0 %	166 %	0 %	0 %
2	Labour Costs	215 %	36 %	749 %	55 %	190 %	626 %	999 %
3	Maintenance Costs	0 %	21 %	0 %	0 %	26 %	0 %	2800 %
4	Cost of Land	-	-	-	-	-	-	-
5	Overhead Costs	0 %	121 %	0 %	0 %	3 %	0 %	220 %
6	Construction Costs	0 %	32 %	0 %	0 %	202 %	89 %	73 %
7	Costs per Order	0 %	38 %	0 %	0 %	33 %	156 %	764 %
8	Transportation Costs	0 %	57 %	7 %	9 %	0 %	168 %	170 %
9	Renewable Energy Sources	0 %	0 %	0 %	0 %	0 %	0 %	0
10	Daylight Usage	0 %	11 %	222 %	0 %	0 %	0 %	169 %
11	Noise Pollution	178 %	87 %	50 %	51 %	89 %	391 %	165 %
12	Recycling Facility	0 %	0 %	0 %	0 %	0 %	0 %	0 %
13	Material Handling Equipment Power Sources	0 %	0 %	0 %	0 %	0 %	0 %	0 %
14	Artificial Lighting Scheme	0 %	33 %	314 %	399 %	7 %	194 %	1454 %
15	Aisle Design	-	-	-	-	-	-	-
16	Work-Life Balance	-	-	-	-	-	-	-
17	Shift Roster	0 %	0 %	0 %	0 %	0 %	0 %	0 %
18	Driver/Operator Training	167 %	1150 %	225 %	0 %	54 %	0 %	180 %
19	Occupational Health and Safety	-	-	-	-	-	-	-
20	Picking Processes	78 %	435 %	235 %	109 %	173 %	362 %	259 %
21	Welfare Facilities	0 %	88 %	0 %	0 %	3 %	0 %	0 %
22	Emergency Room	0 %	18 %	19 %	0 %	0 %	0 %	0 %

 Table 7: Optimal percentage improvements (cost reduction)

5. Discussion and Practical Implications

The study's proposed methodology (AHPSort-traffic light visualization with postoptimal analysis) is general in nature and can be implemented in any company by changing the identified criteria. In the case of XYZ Company Ltd. in this study, experts and decision-makers validated the criteria categorized from the literature review from a triplebottom-line perspective to propose a sustainable criteria framework. This framework, together with the AHPSort-traffic light visualization method, was utilized in XYZ Company Ltd. to evaluate and classify their seven warehouses. The traffic light approach, as mentioned in Figure 2, shows the classification of the warehouses tudied in terms of economic, environmental, and social impact. The most economical warehouses are Warehouses 3 and 4, since almost all their sub-criteria are well within the range as defined by the decision-makers. Similarly, Warehouse 6 is bad, or not economical, since all its subcriteria are in the bad and medium ranges as defined by the experts. Environmentally, warehouses 1, 2, 3, and 4 are good. Warehouses 5, 6, and 7, however, qualify environmentally as medium. Lastly, in terms of social practices, Warehouses 1, 5, 6, and 7 perform well socially, while Warehouse 2 performs worst. In terms of overall sustainability (Table 5), Warehouses 1, 2, 3, 4, and 7 qualify as "good" warehouses, whereas warehouses 5 and 6 qualify as medium warehouses. This classification of warehouses using the proposed framework will help managers and decision-makers strategize to achieve more sustainable warehouse operations. This will also help them categorize the warehouses in terms of all dimensions of sustainability. Post-optimal analysis indicates where significant cost savings already exist, and guides the case company as to what can be improved.

The proposed framework was used in collaboration with the AHPSort-traffic light visualization technique and post-optimal analysis to aid the managerial decision-making process for categorizing warehouses and providing areas for improvement. Overall, the managers and decision-makers now have a clear vision of the sustainability of their warehouses. This will enable them to adopt best practices in the operations of their low-sustainability warehouses to then improve and achieve greater sustainability performance. Moreover, a long-term improvement plan such as transitioning from the consumption of non-renewable to renewable resources (Zailani et al., 2012) will help them achieve excellence in all warehouses. The proposed framework will also help them classify their warehouses in terms of sustainable investment levels. The traffic light approach to warehouse classification will provide the decision-makers with an overview, and indicate

performance in terms of excellent, good, medium, and bad. Finally, the post-optimal analysis results will provide them with indications about where to improve.

6. Conclusion and Future Research

In order to achieve sustainability in their businesses and operations, companies need to transition from traditional ways of doing business to more sustainable ones. Warehouse operations have enjoyed only limited attention in terms of sustainability transitioning. However, warehousing is a major contributor to increasing greenhouse emissions in supply chains (Bartolini, Bottani, and Grosse, 2019). This means that warehousing is one of the major functions of supply chains that needs greater attention when focusing on sustainability in supply chain operations. This study attempted to address this research gap by proposing and evaluating a sustainable criteria framework for warehousing operations in a case company, referred to as XYZ Company Ltd., and based in India. The evaluation was aided by AHPSort traffic light visualization and postoptimal analysis method. The result in Table 5 shows Warehouses 1, 2, 3, 4, and 7 are classified as "good" warehouses whereas Warehouses 6 and 7 are classified as "medium" warehouses in terms of overall sustainability. The post-optimal analysis results indicated certain cost-efficient areas. It also indicated where warehouses require further sustainability strategies to aid in transitioning towards excellence in warehouse performance.

The results of the study are not without limitation, which serve as fertile ground for further research. Future studies can benefit from our proposed framework in the following ways:

 The warehouse sustainability dimensions criteria were identified in the literature review and validated by the case company managers. However, a survey from different companies within a similar sector could be perform to validate the criteria.

- The case company managers set the (limiting) criteria for the profiles. However, a survey of different companies in a similar sector could be performed to validate this limiting profile.
- We implemented our framework in a power plant components company. However, in order to achieve external validity, we would recommend implementing it in other manufacturing or service sectors after slight modifications.

As can be seen, even though this study provides some significant contributions to the academic literature and practice, it sets the stage for additional work on this important sustainable development topic.

References

- Abdallah, T., Diabat, A., & Simchi-Levi, D. (2012). Sustainable supply chain design: a closed-loop formulation and sensitivity analysis. *Production Planning & Control*, *23*(2-3), 120-133. <u>https://doi.org/10.1080/09537287.2011.591622</u>
- Aguado, S., Alvarez, R., & Domingo, R. (2013). Model of efficient and sustainable improvements in a lean production system through processes of environmental innovation. *Journal of Cleaner Production*, 47, 141-148. <u>https://doi.org/10.1016/j.jclepro.2012.11.048</u>
- Aguezzoul, A. (2014). Third-party logistics selection problem: A literature review on criteria and methods. *Omega*, 49, 69-78. <u>https://doi.org/10.1016/j.omega.2014.05.009</u>
- Akarte, M. M., Surendra, N. V., Ravi, B., & Rangaraj, N. (2001). Web based casting supplier evaluation using analytical hierarchy process. *Journal of the Operational Research Society*, 52(5), 511-522. <u>https://doi.org/10.1057/palgrave.jors.2601124</u>
- Almeida-Dias, J., Figueira, J. R., & Roy, B. (2010). Electre Tri-C: A multiple criteria sorting method based on characteristic reference actions. *European Journal of Operational Research*, 204(3), 565-580. <u>https://doi.org/10.1016/j.ejor.2009.10.018</u>
- Amjed, T. W., & Harrison, N. J. (2013). A Model for sustainable warehousing: from theory to best practices. In Proceedings of the International Decision Sciences Institute and Asia Pacific DSI Conference (pp. 1-28).

http://www.conferencemgt.com/conf/idsi13/www/abs130212001.htm

- Ashrafzadeh, M., Rafiei, F. M., Isfahani, N. M., & Zare, Z. (2012). Application of fuzzy TOPSIS method for the selection of Warehouse Location: A Case Study. *Interdisciplinary Journal of Contemporary Research in Business*, *3*(9), 655-671.
- Australia, S. W. (2011). Code of Practice: Managing the work environment and facilities. *Safe Work Australia*. Assessed on 18th January, 2019. <u>https://www.safework.nsw.gov.au/__data/assets/pdf_file/0016/50074/Managing-the-</u> work-environment-and-facilities-COP.pdf

- Badri Ahmadi, H., Kusi-Sarpong, S., & Rezaei, J. (2017). Assessing the social sustainability of supply chains using Best Worst Method. *Resources, Conservation and Recycling, 126*, 99-106. https://doi.org/10.1016/j.resconrec.2017.07.020
- Bai, C., Kusi-Sarpong, S., Badri Ahmadi, H., & Sarkis, J. (2019). Social sustainable supplier evaluation and selection: a group decision-support approach. *International Journal of Production Research*, 57(22), 7046-7067. <u>https://doi.org/10.1080/00207543.2019.1574042</u>
- Bai, C., Kusi-Sarpong, S., & Sarkis, J. (2017). An implementation path for green information technology systems in the Ghanaian mining industry. *Journal of Cleaner Production*, 164, 1105-1123. <u>https://doi.org/10.1016/j.jclepro.2017.05.151</u>
- Barron, H., & Schmidt, C. P. (1988). Sensitivity analysis of additive multiattribute value models. *Operations Research*, *36*(1), 122-127. <u>https://doi.org/10.1287/opre.36.1.122</u>
- BBVA Annual Report (2012). BBVA EAGLEs Emerging and Growth Leading Economies [online]. <u>https://www.bbvaresearch.com/KETD/fbin/mult/120215_BBVAEAGLES_Annual_Report_tcm3</u> <u>48-288784.pdf?ts=1642012</u> (Assessed: 03 April 2019).
- Beise, M., & Rennings, K. (2005). Lead markets and regulation: a framework for analyzing the international diffusion of environmental innovations. *Ecological economics*, *52*(1), 5-17. https://doi.org/10.1016/j.ecolecon.2004.06.007
- Bilbao-Terol, A., Arenas-Parra, M., Cañal-Fernández, V., & Obam-Eyang, P. N. (2018). Multi-criteria analysis of the GRI sustainability reports: an application to Socially Responsible Investment. *Journal of the Operational Research Society*, 69(10), 1576-1598. <u>https://doi.org/10.1057/s41274-017-0229-0</u>
- Buckley, H. (2006). Stop thirsty tyres, presentation to RHA Scotland and Northern Ireland Annual Conference.
- Büyüközkan, G., & Uztürk, D. (2017). Combined QFD TOPSIS approach with 2-tuple linguistic information for warehouse selection. In 2017 IEEE international conference on fuzzy systems (FUZZ-IEEE) (pp. 1-6). IEEE. <u>10.1109/FUZZ-IEEE.2017.8015684</u>
- Cagliano, A. C., DeMarco, A., Rafele, C., & Volpe, S. (2011). Using system dynamics in warehouse management: a fast-fashion case study. *Journal of Manufacturing Technology Management*. 22(2),171-188. <u>https://doi.org/10.1108/17410381111102207</u>
- Cai, W. G., & Zhou, X. L. (2014). On the drivers of eco-innovation: empirical evidence from China. *Journal of Cleaner Production*, *79*, 239-248. https://doi.org/10.1016/j.jclepro.2014.05.035
- CarbonTrust. (2007). Lighting: Bright ideas for more efficient illumination. CTV021, HMSO, London.
- Carter, C. R., & Rogers, D. S. (2008). A framework of sustainable supply chain management: moving toward new theory. *International journal of physical distribution & logistics management, 38*(5), 360-387. <u>https://doi.org/10.1108/09600030810882816</u>
- Cheaitou, A., & Khan, S. A. (2015). An integrated supplier selection and procurement planning model using product predesign and operational criteria. *International Journal on Interactive Design and Manufacturing (IJIDeM)*, 9(3), 213-224. <u>https://doi.org/10.1007/s12008-015-0280-5</u>
- Ciomek, K., Ferretti, V., & Kadziński, M. (2018). Predictive analytics and disused railways requalification: Insights from a Post Factum Analysis perspective. *Decision Support Systems*, *105*, 34-51. <u>https://doi.org/10.1016/j.dss.2017.10.010</u>

- Conceição, S. V., Pedrosa, L. H. P., Neto, A. S. C., Vinagre, M., & Wolff, E. (2012). The facility location problem in the steel industry: a case study in Latin America. *Production Planning & Control*, *23*(1), 26-46. <u>https://doi.org/10.1080/09537287.2010.539991</u>
- Dashore, K., & Sohani, N. (2013). Green supply chain management-barriers & drivers: a review. *International Journal of Engineering Research and Technology*, *2*(4), 2021-2030. https://www.ijert.org/green-supply-chain-management-barriers-drivers-a-review
- da Silva, D. D., de Vasconcelos, N. V. C., & Cavalcante, C. A. V. (2015). Multicriteria decision model to support the assignment of storage location of products in a warehouse. *Mathematical Problems in Engineering*, 2015.1-8. https://doi.org/10.1155/2015/481950
- De Marco, A., & Mangano, G. (2011). Relationship between logistic service and maintenance costs of warehouses. *Facilities*, 29(9/10), 411-421. <u>https://doi.org/10.1108/02632771111146323</u>
- Demirel, T., Demirel, N. Ç., & Kahraman, C. (2010). Multi-criteria warehouse location selection using Choquet integral. *Expert Systems with Applications*, *37*(5), 3943-3952. https://doi.org/10.1016/j.eswa.2009.11.022
- Department for Transport (DfT). (2006). Road Freight Statistics 2005, UK Department for Transport. Assessed on March 2019. <u>http://data.cas.manchester.ac.uk/database3/lan%20Longley/transport/reports/Transport%2</u> 0Statistics%202006.pdf
- Dey, B., Bairagi, B., Sarkar, B., & Sanyal, S. K. (2013). A hybrid fuzzy technique for the selection of warehouse location in a supply chain under a utopian environment. *International Journal of Management Science and Engineering Management*, 8(4), 250-261. <u>https://doi.org/10.1080/17509653.2013.825075</u>
- Dey, B., Bairagi, B., Sarkar, B., & Sanyal, S. K. (2016). Warehouse location selection by fuzzy multicriteria decision making methodologies based on subjective and objective criteria. *International Journal of Management Science and Engineering Management*, 11(4), 262-278. <u>https://doi.org/10.1080/17509653.2015.1086964</u>
- Dey, B., Bairagi, B., Sarkar, B., & Sanyal, S. K. (2017). Group heterogeneity in multi member decision making model with an application to warehouse location selection in a supply chain. Computers & Industrial Engineering, 105, 101-122. https://doi.org/10.1016/j.cie.2016.12.025
- Diabat, A., & Govindan, K. (2011). An analysis of the drivers affecting the implementation of green supply chain management. *Resources, conservation and recycling*, *55*(6), 659-667. https://doi.org/10.1016/j.resconrec.2010.12.002
- Drezner, T., & Drezner, Z. (2012). Modelling lost demand in competitive facility location. *Journal of the Operational Research Society*, 63(2), 201-206. <u>https://doi.org/10.1057/jors.2011.10</u>
- Dubey, R., Bag, S., Ali, S. S., & Venkatesh, V. G. (2013). Green purchasing is key to superior performance: an empirical study. *International Journal of Procurement Management*, 6(2), 187-210. <u>https://doi.org/10.1504/IJPM.2013.052469</u>
- Dutta, B., Singha, T., Goh, M., Lamata, M. T., & Verdegay, J. L. (2019). Post factum analysis in TOPSIS based decision making method. *Expert Systems with Applications*, *138*, 112806. https://doi.org/10.1016/j.eswa.2019.07.023
- Dweiri, F., Khan, S. A., & Almulla, A. (2018). A multi-criteria decision support system to rank sustainable desalination plant location criteria. *Desalination*, 444, 26-34. https://doi.org/10.1016/j.desal.2018.07.007

- Dweiri, F., Khan, S. A., & Jain, V. (2015). Production planning forecasting method selection in a supply chain: a case study. *International Journal of Applied Management Science*, 7(1), 38-58. https://doi.org/10.1504/IJAMS.2015.068056
- Dweiri, F., Kumar, S., Khan, S. A., & Jain, V. (2016). Designing an integrated AHP based decision support system for supplier selection in automotive industry. *Expert Systems with Applications*, 62, 273-283. https://doi.org/10.1016/j.eswa.2016.06.030
- European Logistics Association and Kearney, A. and others. (2004). *Differentiation for Performance* - *Excellence in Logistics 2004. Deutscher Verkehrs-Verlag, Hamburg.*
- Fahimnia, B., Sarkis, J., & Eshragh, A. (2015). A tradeoff model for green supply chain planning: A leanness-versus-greenness analysis. *Omega*, 54, 173-190. https://doi.org/10.1016/j.omega.2015.01.014
- Foroozesh, N., Tavakkoli-Moghaddam, R., & Mousavi, S. M. (2018). A novel group decision model based on mean–variance–skewness concepts and interval-valued fuzzy sets for a selection problem of the sustainable warehouse location under uncertainty. *Neural Computing and Applications*, 30(11), 3277-3293. <u>https://doi.org/10.1007/s00521-017-2885-z</u>
- Fragoso, R., & Figueira, J. R. (2020). Sustainable supply chain network design: An application to the wine industry in Southern Portugal. *Journal of the Operational Research Society*, advance online publication. <u>https://doi.org/10.1080/01605682.2020.1718015</u>
- Neto, J. Q. F., Bloemhof-Ruwaard, J. M., van Nunen, J. A., & van Heck, E. (2008). Designing and evaluating sustainable logistics networks. *International Journal of Production Economics*, *111*(2), 195-208. <u>https://doi.org/10.1016/j.ijpe.2006.10.014</u>
- Gallmann, F., & Belvedere, V. (2011). Linking service level, inventory management and warehousing practices: A case-based managerial analysis. *Operations Management Research*, *4*(1-2), 28-38. <u>https://doi.org/10.1007/s12063-010-0043-1</u>
- García, J. L., Alvarado, A., Blanco, J., Jiménez, E., Maldonado, A. A., & Cortés, G. (2014). Multiattribute evaluation and selection of sites for agricultural product warehouses based on an analytic hierarchy process. *Computers and Electronics in Agriculture*, 100, 60-69. <u>https://doi.org/10.1016/j.compag.2013.10.009</u>
- García-Dastugue, S., & Eroglu, C. (2019). Operating performance effects of service quality and environmental sustainability capabilities in logistics. *Journal of Supply Chain Management*, 55(3), 68-87. <u>https://doi.org/10.1111/jscm.12185</u>
- Garetti, M., & Taisch, M. (2012). Sustainable manufacturing: trends and research challenges. *Production planning & control*, *23*(2-3), 83-104. <u>https://doi.org/10.1080/09537287.2011.591619</u>
- Gazeley. (2004). Eco template a framework for increasingly environmental and socially responsible logistics development. Gazeley UK Limited, London.
- Glasmeier, A. K. K.-J. (1996). Power Shift The Rising Control of Distributors and Retailers in the Supply Chain for Manufactured Goods. *Urban Geography*. 17(8), 740-757. https://doi.org/10.2747/0272-3638.17.8.740
- González, P., Sarkis, J., & Adenso-Díaz, B. (2008). Environmental management system certification and its influence on corporate practices. *International Journal of Operations & Production Management*. 28(11), 1021-1041. <u>https://doi.org/10.1108/01443570810910179</u>
- Govindan, K., Kadziński, M., Ehling, R., & Miebs, G. (2019). Selection of a sustainable third-party reverse logistics provider based on the robustness analysis of an outranking graph kernel

conducted with ELECTRE I and SMAA. *Omega*, *85*, 1-15. <u>https://doi.org/10.1016/j.omega.2018.05.007</u>

- Gunasekaran, A., Irani, Z., & Papadopoulos, T. (2014). Modelling and analysis of sustainable operations management: certain investigations for research and applications. *Journal of the Operational Research Society*, 65(6), 806-823. <u>https://doi.org/10.1057/jors.2013.171</u>
- Gunasekaran, A., Patel, C., & Tirtiroglu, E. (2001). Performance measures and metrics in a supply chain environment. *International Journal of Operations & Production Management*, *21*(1/2), 71-87. <u>https://doi.org/10.1108/01443570110358468</u>
- Gunasekaran, A., Marri, H. B., & Menci, F. (1999). Improving the effectiveness of warehousing operations: a case study. *Industrial Management & Data Systems*, 99(8), 328-339. https://doi.org/10.1108/02635579910291975
- Gupta, H., Kusi-Sarpong, S., & Rezaei, J. (2020). Barriers and overcoming strategies to supply chain sustainability innovation. *Resources, Conservation and Recycling*, *161*, 104819. <u>https://doi.org/10.1016/j.resconrec.2020.104819</u>
- He, Y., Wang, X., Lin, Y., Zhou, F., & Zhou, L. (2017). Sustainable decision making for joint distribution center location choice. *Transportation Research Part D: Transport and Environment*, 55, 202-216. <u>https://doi.org/10.1016/j.trd.2017.07.001</u>
- Hosseini-Motlagh, S. M., Ebrahimi, S., & Jokar, A. (2019). Sustainable supply chain coordination under competition and green effort scheme. *Journal of the Operational Research Society*, advance online publication. <u>https://doi.org/10.1080/01605682.2019.1671152</u>
- Hudson. (2005). The Case for Work–Life Balance: Closing the gap between policy and practice. 20: 20 Series, 3-24.
- Hyde, K. M., & Maier, H. R. (2006). Distance-based and stochastic uncertainty analysis for multicriteria decision analysis in Excel using Visual Basic for Applications. *Environmental Modelling* & Software, 21(12), 1695-1710. <u>https://doi.org/10.1016/j.envsoft.2005.08.004</u>
- Hyde, K. M., Maier, H. R., & Colby, C. B. (2005). A distance-based uncertainty analysis approach to multi-criteria decision analysis for water resource decision making. *Journal of Environmental Management*, 77(4), 278-290. <u>https://doi.org/10.1016/j.jenvman.2005.06.011</u>
- Ishizaka, A., Pearman, C., & Nemery, P. (2012). AHPSort: an AHP-based method for sorting problems. *International Journal of Production Research*, *50*(17), 4767-4784. https://doi.org/10.1080/00207543.2012.657966
- Ishizaka, A., & Nemery, P. (2014). Assigning machines to incomparable maintenance strategies with ELECTRE-SORT. *Omega*, 47, 45-59. <u>https://doi.org/10.1016/j.omega.2014.03.006</u>
- Ishtiaq, P., Khan, S. A., & Haq, M. U. (2018). A multi-criteria decision-making approach to rank supplier selection criteria for hospital waste management: A case from Pakistan. *Waste Management & Research*, *36*(4), 386-394. <u>https://doi.org/10.1177/0734242X18755894</u>
- Jain, V., & Khan, S. A. (2017). Application of AHP in reverse logistics service provider selection: a case study. *International Journal of Business Innovation and Research*, *12*(1), 94-119. https://doi.org/10.1504/IJBIR.2017.080711
- Jayant, A., Gupta, P., Garg, S. K., & Khan, M. (2014). TOPSIS-AHP based approach for selection of reverse logistics service provider: a case study of mobile phone industry. *Procedia engineering*, 97, 2147-2156. <u>https://doi.org/10.1016/j.proeng.2014.12.458</u>

- Jha, M. K., Raut, R. D., Gardas, B. B., & Raut, V. (2018). A sustainable warehouse selection: an interpretive structural modelling approach. *International Journal of Procurement Management*, 11(2), 201-232. <u>https://doi.org/10.1504/IJPM.2018.090025</u>
- Ji, G., & Gunasekaran, A. (2014). Evolution of innovation and its strategies: from ecological niche models of supply chain clusters. *Journal of the Operational Research Society*, 65(6), 888-903. https://doi.org/10.1057/jors.2013.136
- Johnson, A., Chen, W. C., & McGinnis, L. F. (2010). Large-scale Internet benchmarking: Technology and application in warehousing operations. *Computers in Industry*, 61(3), 280-286. <u>https://doi.org/10.1016/j.compind.2009.10.006</u>
- Johnson, A., & McGinnis, L. (2010). Performance measurement in the warehousing industry. *IIE Transactions*, 43(3), 220-230. <u>https://doi.org/10.1080/0740817X.2010.491497</u>
- Johnson, E. (2008). Disagreement over carbon footprints: A comparison of electric and LPG forklifts. *Energy Policy*, *36*(4), 1569-1573. <u>https://doi.org/10.1016/j.enpol.2008.01.014</u>
- Kadziński, M., Ciomek, K., Rychły, P., & Słowiński, R. (2016). Post factum analysis for robust multiple criteria ranking and sorting. *Journal of Global Optimization*, 65(3), 531-562. https://doi.org/10.1007/s10898-015-0359-3
- Karmaker, C., & Saha, M. (2015). Optimization of warehouse location through fuzzy multi-criteria decision making methods. *Decision Science Letters*, 4(3), 315-334. <u>https://doi.org/10.5267/j.dsl.2015.4.005</u>
- Ketikidis, P. H., Hayes, O. P., Lazuras, L., Gunasekaran, A., & Koh, S. L. (2013). Environmental practices and performance and their relationships among Kosovo construction companies: a framework for analysis in transition economies. *International Journal of Services and Operations Management*, 14(1), 115-130. <u>https://doi.org/10.1504/IJSOM.2013.050565</u>
- Khan, S. A., Chaabane, A., & Dweiri, F. (2019). A knowledge-based system for overall supply chain performance evaluation: a multi-criteria decision making approach. *Supply Chain Management: An International Journal*, 24(3), 377-396. <u>https://doi.org/10.1108/SCM-06-2017-0197</u>
- Khan, S. A., Chaabane, A., & Dweiri, F. T. (2018). Multi-Criteria Decision-Making Methods Application in Supply Chain Management: A Systematic Literature. In Multi-Criteria Methods and Techniques Applied to Supply Chain Management (Vol. 1). BoD–Books on Demand. <u>https://doi: 10.5772/intechopen.74067</u>
- Khan, S. A., Dweiri, F., & Chaabane, A. (2016, December). Fuzzy-AHP approach for warehouse performance measurement. In 2016 IEEE International Conference on Industrial Engineering and Engineering Management (IEEM) (pp. 871-875). IEEE. https://doi.org/10.1109/IEEM.2016.7798001
- Khan, S. A., Dweiri, F., & Jain, V. (2016). Integrating analytical hierarchy process and quality function deployment in automotive supplier selection. *International Journal of Business Excellence*, 9(2), 156-177. <u>https://doi.org/10.1504/IJBEX.2016.074851</u>
- Khan, S. A., & Hosany, Y. I. A. (2016). Multi-million construction contractor selection: a comparative study. *International Journal of Intelligent Enterprise*, *3*(2), 93-119. https://doi.org/10.1504/IJIE.2016.076030
- Khan, S. A., Kusi-Sarpong, S., Arhin, F. K., & Kusi-Sarpong, H. (2018). Supplier sustainability performance evaluation and selection: A framework and methodology, *Journal of Cleaner*

Production, 205, 964-979. https://doi.org/10.1016/j.jclepro.2018.09.144

Khumawala, B. M., & Whybark, D. C. (1976). Solving the dynamic warehouse location problem. *International Journal of Physical Distribution*, 6(5), 238-251. https://doi.org/10.1108/eb014375

- Kirchoff, J. F., Omar, A., & Fugate, B. S. (2016). A behavioral theory of sustainable supply chain management decision making in non-exemplar firms. *Journal of supply chain management*, 52(1), 41-65. <u>https://doi.org/10.1111/jscm.12098</u>
- Korpela, J., & Tuominen, M. (1996). A decision aid in warehouse site selection. *International Journal of Production Economics*, 45(1-3), 169-180. <u>https://doi.org/10.1016/0925-5273(95)00135-2</u>
- Krejčí, J., & Ishizaka, A. (2018). FAHPSort: A fuzzy extension of the AHPSort method. International Journal of Information Technology & Decision Making, 17(04), 1119-1145. <u>https://doi.org/10.1142/S0219622018400011</u>
- Kuorinka, I., Lortie, M., & Gautreau, M. (1994). Manual handling in warehouses: The illusion of correct working postures. *Ergonomics*, 37(4), 655-61. https://doi.org/10.1080/00140139408963680
- Kusi-Sarpong, S., Gupta, H., Khan, S. A., Jabbour, C. J. C., Rehman, S. T., & Kusi-Sarpong, H. (2019). Sustainable supplier selection based on industry 4.0 initiatives within the context of circular economy implementation in supply chain operations. *Production Planning and Control.* (in press)
- Kusi-Sarpong, S., Gupta, H., & Sarkis, J. (2019). A supply chain sustainability innovation framework and evaluation methodology. *International Journal of Production Research*, 57(7), 1990-2008. https://doi.org/10.1080/00207543.2018.1518607
- Kusi-Sarpong, S., & Sarkis, J. (2019). Sustainable supply chains and emerging economies. *Resources, Conservation and Recycling*, 143, 238-243. <u>https://doi.org/10.1016/j.resconrec.2019.01.016</u>
- Kusi-Sarpong, S., Sarkis, J., & Wang, X. (2016). Assessing green supply chain practices in the Ghanaian mining industry: A framework and evaluation. *International Journal of Production Economics*, *181*, 325-341. <u>https://doi.org/10.1016/j.ijpe.2016.04.002</u>
- Kusi-Sarpong, S., Varela, M. L., Putnik, G., Ávila, P., & Agyemang, J. (2018). Supplier evaluation and selection: A fuzzy novel multi-criteria group decision-making approach. *International Journal* for Quality Research, 12(2), 459–486. <u>https://doi.org/10.18421/IJQR12.02-10</u>
- Lambert, D., Stock, J. R., & Ellram, L. M. (1998). *Fundamentals of logistics management*. McGraw-Hill/Irwin. <u>https://doi.org/10.1007/978-3-540-24816-3_1</u>
- Larsson, T. J., & Rechnitzer, G. (1994). Forklift trucks—analysis of severe and fatal occupational injuries, critical incidents and priorities for prevention. *Safety Science*, *17*(4), 275-289. https://doi.org/10.1016/0925-7535(94)90029-9
- Linton, J. D., Klassen, R., & Jayaraman, V. (2007). Sustainable supply chains: An introduction. *Journal of operations management*, *25*(6), 1075-1082. https://doi.org/10.1016/j.jom.2007.01.012
- Lolli, F., Ishizaka, A., & Gamberini, R. (2014). New AHP-based approaches for multi-criteria inventory classification. *International Journal of Production Economics*, 156, 62-74.

https://doi.org/10.1016/j.ijpe.2014.05.015

- López, C., & Ishizaka, A. (2017). GAHPSort: A new group multi-criteria decision method for sorting a large number of the cloud-based ERP solutions. *Computers in Industry*, 92, 12-25. <u>https://doi.org/10.1016/j.compind.2017.06.007</u>
- Luthra, S., Kumar, V., Kumar, S., & Haleem, A. (2011). Barriers to implement green supply chain management in automobile industry using interpretive structural modeling technique: An Indian perspective. *Journal of Industrial Engineering and Management*, *4*(2), 231-257. doi:10.3926/jiem.v4n2.p231-257
- Luthra, S., Qadri, M. A., Garg, D., & Haleem, A. (2014). Identification of critical success factors to achieve high green supply chain management performances in Indian automobile industry. *International Journal of Logistics Systems and Management*, *18*(2), 170-199. https://doi.org/10.1504/IJLSM.2014.062325
- Marchant, C. (2010) "Reducing the environmental impact of warehousing". Chapter 8, pg 167-192 In McKinnon, A., Cullinane, S., Browne, M. Whiting, A. (Ed.), Green Logistics Improving the environmental sustainability of logistics, Kogan Page, India
- Magee, J.F., Copacino, W.C. and Rosenfield, D.B. (1985) Modern Logistics Management: Integrating Marketing, Manufacturing and Physical Distribution, John Wiley & Sons, New York, USA
- Massey, C., Lamm, F., and Perry, M. (2006). Understanding the Link Between Workplace Health & Safety & Firm Performance & Productivity.Department of Labour, Wellington
- Mehta, Y., & Rajan, A. J. (2017). Manufacturing sectors in India: outlook and challenges. *Procedia* engineering, 174, 90-104. <u>https://doi.org/10.1016/j.proeng.2017.01.173</u>
- Meneghetti, A., & Monti, L. (2013). Sustainable storage assignment and dwell-point policies for automated storage and retrieval systems. *Production Planning & Control, 24*(6), 511-520. https://doi.org/10.1080/09537287.2011.637525
- Miccoli, F., & Ishizaka, A. (2017). Sorting municipalities in Umbria according to the risk of wolf attacks with AHPSort II. *Ecological Indicators*, 73, 741-755. <u>https://doi.org/10.1016/j.ecolind.2016.10.034</u>
- Michel, R. (2014). Why sustainable design still matters. *Logistics management (Highlands Ranch, Colo.: 2002)*, 53(2), 48-52.
- Millet, I. (1997). The effectiveness of alternative preference elicitation methods in the analytic hierarchy process. *Journal of Multi-Criteria Decision Analysis*, 6(1), 41-51. https://doi.org/10.1002/(SICI)1099-1360(199701)6:1<41::AID-MCDA122>3.0.CO;2-D
- Mohsen, & Hassan, M. D. (2002). A framework for the design of warehouse layout. *Facilities*, *20*(13/14), 432-440. <u>https://doi.org/10.1108/02632770210454377</u>
- National Occupational Health and Safety Advisory Committee (NOHSAC). (2006). The Economic and Social Costs of Occupational Disease and Injury in New Zealand. NOHSAC Technical Report Number 4. Access Economics, Wellington. Assessed on 15th Febuary, 2019. <u>http://psm-dm.otago.ac.nz/ipru/ReportsPDFs/OR058.pdf</u>
- Nemery, P., & Lamboray, C. (2008). FlowSort: a flow-based sorting method with limiting or central profiles, TOP ,16, 90-113. <u>https://doi.org/10.1007/s11750-007-0036-x</u>

- New Zealand Department of Labour (NZDoL), (2007). Managing Shift Work to Minimise Workplace Fatigue. Department of Labour, Wellington. Assessed on 15th Febuary, 2019. <u>https://worksafe.govt.nz/topic-and-industry/work-related-health/fatigue/fatigue-quick-guide/</u>
- NZBCSD (2003). Business Guide to a Sustainable Supply Chain: A Practical Guide. New Zealand Business Council for Sustainable Development. Assessed on 20th Febuary, 2019. <u>https://www.sbc.org.nz/_data/assets/pdf_file/0005/54914/Sustainable-Supply-Chain-Guide.pdf</u>
- Orji, I. J., Kusi-Sarpong, S., Gupta, H., & Okwu, M. (2019). Evaluating challenges to implementing eco-innovation for freight logistics sustainability in Nigeria. *Transportation Research Part A: Policy and Practice*, *129*, 288-305. <u>https://doi.org/10.1016/j.tra.2019.09.001</u>
- Orji, I. J., Kusi-Sarpong, S., Huang, S., & Vazquez-Brust, D. (2020). Evaluating the factors that influence blockchain adoption in the freight logistics industry. *Transportation Research Part E: Logistics and Transportation Review*, *141*, 102025. https://doi.org/10.1016/j.tre.2020.102025
- Özcan, T., Çelebi, N., & Esnaf, Ş. (2011). Comparative analysis of multi-criteria decision making methodologies and implementation of a warehouse location selection problem. *Expert Systems with Applications*, *38*(8), 9773-9779. <u>https://doi.org/10.1016/j.eswa.2011.02.022</u>
- Pagell, M., & Shevchenko, A. (2014). Why research in sustainable supply chain management should have no future. *Journal of supply chain management*, *50*(1), 44-55. <u>https://doi.org/10.1111/jscm.12037</u>
- Perrott, B. (2014). The sustainable organisation: Blueprint for an integrated model. *Journal of Business Strategy*, 35(3), 26-37. <u>https://doi.org/10.1108/JBS-07-2013-0061</u>
- Por, H. H., & Budescu, D. V. (2017). Eliciting Subjective Probabilities through Pair-wise Comparisons. *Journal of Behavioral Decision Making*, 30(2), 181-196. <u>https://doi.org/10.1002/bdm.1929</u>
- Ramaa, A., Subramanya, K. N., & Rangaswamy, T. M. (2012). Impact of warehouse management system in a supply chain. *International Journal of Computer Applications*, 54(1), 14-20. https://doi.org/10.5120/8530-2062
- Rao, C., Goh, M., Zhao, Y., & Zheng, J. (2015). Location selection of city logistics centers under sustainability. *Transportation Research Part D: Transport and Environment*, 36, 29-44. <u>https://doi.org/10.1016/j.trd.2015.02.008</u>
- Rao, P. (2002). Greening the supply chain: a new initiative in South East Asia. International Journal of Operations & Production Management, 22(6), 632-655. https://doi.org/10.1108/01443570210427668
- Raut, R. D., Narkhede, B. E., Gardas, B. B., & Raut, V. (2017). Multi-criteria decision making approach: a sustainable warehouse location selection problem. *International Journal of Management Concepts and Philosophy*, 10(3), 260-281. https://doi.org/10.1504/IJMCP.2017.085834
- Rehman, S. T., Khan, S. A., Kusi-Sarpong, S., & Hassan, S. M. (2018). Supply chain performance measurement and improvement system. *Journal of Modelling in Management*, *13*(3), 522-549 <u>https://doi.org/10.1108/JM2-02-2018-0012</u>
- Rimiene, K. (2008). The design and operation of warehouse. Economics & Management, 13, 136-

137.

- Roh, S. Y., Jang, H. M., & Han, C. H. (2013). Warehouse location decision factors in humanitarian relief logistics. *The Asian Journal of Shipping and Logistics*, *29*(1), 103-120. https://doi.org/10.1016/j.ajsl.2013.05.006
- Saaty, T. L. (1977). A scaling method for priorities in hierarchical structures. *Journal of mathematical psychology*, *15*(3), 234-281. <u>https://doi.org/10.1016/0022-2496(77)90033-5</u>
- Saaty, T. L. (1980). The Analytic Hierarchy Process. McGraw-Hill, New York.
- Saaty, T. L. (2003). Decision-making with the AHP: Why is the Principal Eigenvector necessary? *European Journal of Operational Research*, 145, 85-91. <u>https://doi.org/10.1016/S0377-</u> <u>2217(02)00227-8</u>
- Sarkis, J. (2018). Sustainable and green supply chains: Advancement through Resources, Conservation and Recycling. *Resources, Conservation and Recycling*, 134, A1-A3. <u>https://doi.org/10.1016/j.resconrec.2017.12.022</u>
- Schmitt, A. J., Sun, S. A., Snyder, L. V., & Shen, Z. J. M. (2015). Centralization versus decentralization: Risk pooling, risk diversification, and supply chain disruptions. *Omega*, 52, 201-212. <u>https://doi.org/10.1016/j.omega.2014.06.002</u>
- Sivitanidou, R. (1996). Warehouse and distribution facilities and community attributes: an empirical study. *Environment and Planning A*, *28*(7), 1261-1278. <u>https://doi.org/10.1068/a281261</u>
- Stolze, H., Mollenkopf, D., Thornton, L., Brusco, M. & Flint, D. (2018). Supply Chain and Marketing Integration: Tension in Frontline Social Networks. *Journal of Supply Chain Management*, 54(3), 3-21. <u>https://doi.org/10.1111/jscm.12169</u>
- Song, Y., & Li, G. (2019). A large-scale group decision-making with incomplete multi-granular probabilistic linguistic term sets and its application in sustainable supplier selection. *Journal* of the Operational Research Society, 70(5), 827-841. https://doi.org/10.1080/01605682.2018.1458017
- Tan, K. S., Ahmed, M. D., & Sundaram, D. (2010). Sustainable enterprise modelling and simulation in a warehousing context. *Business Process Management Journal*, 16(5), 871 - 886. <u>https://doi.org/10.1108/14637151011076511</u>
- Tompkins, J. A., & Smith, J. D. (1998). *The Warehouse Management Handbook*. Tompkins Press, Raleigh, North Carolina.
- Tompkins, J. A., White, J. A., Bozer, Y. A., Frazelle, E. H., Tanchoco, J. M. A., and Trevino, J. (1996). Facilities Planning. 2nd Edition, John Wiley & Sons, New York
- Triantaphyllou, E., & Sánchez, A. (1997). A sensitivity analysis approach for some deterministic multi-criteria decision-making methods. *Decision sciences*, *28*(1), 151-194. https://doi.org/10.1111/j.1540-5915.1997.tb01306.x
- Turğut, B. T., Taş, G., Herekoğlu, A., Tozan, H., & Vayvay, O. (2011). A fuzzy AHP based decision support system for disaster center location selection and a case study for Istanbul. *Disaster Prevention and Management: An International Journal*, 20(5), 499-520. https://doi.org/10.1108/09653561111178943
- Uysal, F., & Tosun, Ö. (2014). Selection of sustainable warehouse location in supply chain using the grey approach. *International Journal of Information and Decision Sciences*, 6(4), 338-353. https://doi.org/10.1504/IJIDS.2014.066633
- Varadarajan, R. (2017). Innovating for sustainability: a framework for sustainable innovations and

a model of sustainable innovations orientation. *Journal of the Academy of Marketing Science*, 45(1), 14-36. <u>https://doi.org/10.1007/s11747-015-0461-6</u>

- Victoria, W. (2008). Workplace amenities and work environment. WorkSafe Victoria. Assessed on 25th March, 2019. <u>https://www.worksafe.vic.gov.au/resources/compliance-code-workplace-amenities-and-work-environment</u>
- Ward, D., Tyler, P., Wilson, P. and Eichinger, M., Pat, W. (2004). Developments in Rail Simulators and Computer Based Training to Increase Training. In CORE 2000: Railway Technology for the 21st Century. 229-235. <u>https://trove.nla.gov.au/work/39448015</u>
- Developments in Rail Simulators and Computer Based Training to Increase Training Efficiency and Effectiveness. Simulation Industry Association of Australia, Lindfield
- Wolters, W. T. M., & Mareschal, B. (1995). Novel types of sensitivity analysis for additive MCDM methods. *European Journal of Operational Research*, *81*(2), 281-290. https://doi.org/10.1016/0377-2217(93)E0343-V
- Wruck, S., Vis, I. F., & Boter, J. (2013). Time-restricted batching models and solution approaches for integrated forward and return product flow handling in warehouses. *Journal of the Operational Research Society*, 64(10), 1505-1516. <u>https://doi.org/10.1057/jors.2012.137</u>
- Wu, P., Jin, Y., Shi, Y., & Shyu, H. (2017). The impact of carbon emission costs on manufacturers' production and location decision. *International Journal of Production Economics*, 193, 193-206. <u>https://doi.org/10.1016/j.ijpe.2017.07.005</u>
- Wutthisirisart, P., Sir, M. Y., & Noble, J. S. (2015). The two-warehouse material location selection problem. *International Journal of Production Economics*, *170*, 780-789. <u>https://doi.org/10.1016/j.ijpe.2015.07.008</u>
- Xiao, C., Wilhelm, M., van der Vaart, T., & van Donk, D. P. (2019). Inside the buying firm: Exploring responses to paradoxical tensions in sustainable supply chain management. *Journal of Supply Chain Management*, *55*(1), 3-20. <u>https://doi.org/10.1111/jscm.12170</u>
- Xing, Y., & Dangerfield, B. (2011). Modelling the sustainability of mass tourism in island tourist economies. *Journal of the Operational Research Society*, *62*(9), 1742-1752. <u>https://doi.org/10.1057/jors.2010.77</u>
- Xu, C., Zhao, D., Min, J., & Hao, J. (2020). An inventory model for nonperishable items with warehouse mode selection and partial backlogging under trapezoidal-type demand. *Journal* of the Operational Research Society, advance online publication. <u>https://doi.org/10.1080/01605682.2019.1708822</u>
- Yuan, Q. (2018). Environmental justice in warehousing location: State of the art. *Journal of Planning Literature*, *33*(3), 287-298. <u>https://doi.org/10.1177/0885412217753841</u>
- Zailani, S., Jeyaraman, K., Vengadasan, G., & Premkumar, R. (2012). Sustainable supply chain management (SSCM) in Malaysia: A survey. *International Journal of Production Economics*, *140*(1), 330-340. <u>https://doi.org/10.1016/j.ijpe.2012.02.008</u>
- Żuchowski, W. (2015). Division of environmentally sustainable solutions in warehouse management and example methods of their evaluation. *LogForum*, *11*(2), 171-182. <u>https://doi.org/10.17270/J.LOG.2015.2.5</u>